Book Order

ffl The bookstore seems to have screwed up the book order for the course.

– Moore: Concepts and Controversies is out of stock.

We expect the bookstore to be restocked by the end of the week. Please send email if you cannot get the book by then.

– Siegel and Morgan: Statistics and Data analysis is in stock.

– Minitab Handbook is in stock.

HW 1 Excused

ffl Since HW 1 is all out of the Moore book,

Late Homework will not be penalized this week only, as long as it is in my office (BH 232 C) by 5:00 Mon Jan 25, 1999.

0-0